Intelligent Computing Techniques for the Detection of Sleep Disorders: A Review

Vijay Kumar Garg Research Scholar School of Computer Applications Guru Kashi University

ABSTRACT

Intelligent computing methods and knowledge based systems are well known techniques used for the detection of various medical disorders. This paper is based on the review of various intelligent computing methods that are used to detect sleep disorders. The main concern is based on the detection of sleep disorders such as sleep apnea, insomnia, parasomnia and snoring. The most common diagnostic methods used by many researchers are based on knowledge-based system (KBS), rule based reasoning (RBR), case based reasoning (CBR), fuzzy logic (FL), artificial neural network (ANN), support vector machine(SVM), multi-layer perceptron (MLP) neural network, genetic algorithm (GA), k-nearest neighbor (k-NN), hybrid neural network, bayesian network (BN), data mining (DM) and many other integrated approaches. In traditional approach questionnaire was used for the detection of various disorders that is now overcome with all above mentioned techniques to enhance the accuracy, sensitivity and specificity.

Keywords

RBR, CBR, ANN, GA, DM, FL, BN, Intelligent computing techniques

1. INTRODUCTION

Sleep plays a vital position in the tradition of neuroscience. The subject area of sleep disorders becomes very important due to their familiarity in universe (Stanley J. Swierzewski, 2011). In 1987, Klink studied that 41% of all considered subjects had at least one syndrome of disrupted sleep (Krajewski, 2007). In 2004, Young narrated that one out of 5 adults is distressed from daytime sleepiness (Jerome, 1998). Sleep apnea (Guimaraes, 2001) and narcolepsy are two problems raised due to extreme daylight sleep. These two disorders constitute a confusing or some time vital effect on regular activities. Likely, the disturbances in sleep are created by a sleep-related breathing disorder known as sleep apnea (Liu, 2008). Also, sleep disorders have many concise and long duration dreadful effects. The short living effect directs to impaired attention, impact on quality of life, less potency and chances of mishap increasing. The long-term effect of sleep distressed move towards incrementing the morbidity and mortality rate from the increasing mishaps, cardiovascular diseases, high blood pressure, bulkiness and learning disability along with discouragement (Maali, 2012). Some sleep disorders are severe enough to hamper the physical, psychological, cognitive and motor functioning of an individual. In daily life, we usually pay no attention to snoring, sleep apnea, insomnia, parasomnia, but there persistence can be serious. Therefore, early detection of sleep disorder becomes prime task. In early stage researchers uses the quantitative approach in the form of questionnaire for the detection of

R.K. Bansal Professor and Dean Research School of Computer Applications Guru Kashi University

various sleep disorders. But accuracy measurement is always a problem in this approach because whole questionnaire survey is dependent on total number of participants as well as questions designed for the survey. So, the researchers move towards to enhance the accuracy by using various intelligent techniques.

Artificial Intelligence plays a vital role in this direction. Intelligent computing techniques such as: ANN, FL, GA, SVM, MLPNN, DM, BN, all are data controlling instead of knowledge. But now many researchers have adopted the various integration technique of knowledge controlling that is applicable in medical domain. The basic problemsolving paradigms in the field of artificial intelligence are RBR and CBR (Pandey, 2009). Rechtschaffen and Kales (R&K) recommended a rigid standard for specification of patient's sleep macrostructure which is recently mutated by the American Academy of Sleep Medicine (Estevez, 2012). The human scoring is tedious and extravagant. Hence, frequent attempts have been done to create such systems that count the records automatically (Maali, 2012). The R&K criteria used to construct the rule based sleep staging system by using multi-rule decision tree. Multiple decision tree approach was used to improve the accuracy over a single decision tree and it was found that multiple decision tree has 7% more accuracy over a single model (Khasawneh, 2012). RBR methods are also time consuming methods as they require signal information, detection of specific patterns like K-complexes, sleep spindles in EEG and rapid eye movements in EOG. Canisius used the bio-signals processing algorithms for the detection of sleep disorders breathing from ECG signals with the accuracy of 77% (Canisius, 2008). Much time is required to create a system from RBR which extract features from original information to construct the rules as per human brain while in opposite numerical classification doesn't need any rule base and features are extracted by power spectrum (Tian, 2005). The chromosomes having variable length structure and fitness function are used to find the optimal input features and recognition of network specifications (Kim, 2000). In numerical classification method, no such human knowledge and rules are required (Tian, 2005). The advancement of a portable microcontroller based device can be better proved for longterm and home monitoring of snoring. This type of device can provide various outputs as: total snoring count, medium number of snores per hour, and the number of irregular snoring. The success rate of this device was 85% in a lab environment and around 70% in a home (Hsu, 2005). CBR is able to utilize the exact knowledge of previously experienced, concrete problem cases. It favors learning from experience, as it is easier to learn from an actual problem solving experience (Lopez, 1997 and Khan, 2003). The first system that might be called a case-based reasoned was the CYRUS system, developed by Janet

Kolodner. CYRUS was a demonstration of Schank's dynamic memory model (Ian Watson and Farhi Marir, 2013). Due to contrasted advantage and disadvantage of RBR and CBR, sometimes it is difficult to solve problem independently. But, if their advantages are realized and disadvantages are eliminated then their different junctions presents considerable advantages like as BOLERO (Lopez, 1997) and MIKAS (Khan, 2003) which integrate RBR and CBR, PROTOS (Porter, 1986) and CASEY (Koton, 1988) which integrate CBR and MBR, T-IDDM (Montani, 2003) which integrate RBR, CBR and MBR and GREBE that has knowledge base consists of rule and cases related with laws for injuries to workers (Aamodt, 1994 and Lopez, 1997). The modification of hidden knowledge into precise rules would lead to loss and irregularity of knowledgeable contents (Zurada, 1992). An Alternative of such type of inference is baye's theorem that fits a probabilistic value for every measured output like as ES (Bernstein, 1995) and MES (Dragulescu, 2007 and Vinterbo, 2000). SAMOA (Mariano, 2004) is a automatic sleep apnea syndrome diagnostic system. All these systems are successful for special diseases and self-governing symptoms but fails when a person can have more than one disease and symptoms of the some diseases have same reason to grow up. Also rule-based expert system has two limitations: 1) all conditions cannot be explained by rules for various situations and 2) experiences gathered by trial and errors cannot be easily contained to the knowledge base without human effort, due to this it has low agreement rate 83.0% (Park, 2000). Therefore, ANN has entertained to implement the human intelligence. ANN has been widely used and accepted method for the diagnosis of sleep stages and its various disorders.

In some cases, GA is used to determine the number of neurons of the hidden layer. There is always a difficulty in the development of an automated system due to some uncertainties which arises as the problems increased day by day. To handle this kind of situation, FL is measured as a suitable tool like as ISSSC version 1.0 system which is helpful for medical diagnosis (Pinero, 2004). Data mining (DM) is a proficient tool and technique used for the formation of new knowledge from databases. Various techniques of DM are considered in the diagnosis of many diseases like as prediction rule for obstructive sleep apnea (Kwiatkowska, 2007).

This paper is based on the review of the different methods of detection and diagnosis of sleep diseases such as sleep apnea, insomnia, parasomnia, snoring. We have covered the methods based upon many intelligent computing techniques (ICT) and their combinations such as KBS/ RBR, CBR, ANN, GA, FL, DM. The combined methods are RBR-CBR, CBR-FL, RBR-ANN, ANN-FL, ANN-BN, ANN-GA, ANN-DM, GA-FL. Data mining methods and Expert systems (Bertha, 2012 and Yldiz, 2011) proposed for the detection of sleep disorders in have also been described in this work. Wireless technology is also playing a key role in sleep studies where many persons can acquire the advantages of diagnosis and treatment, even without creating any disturbances in patient's normal sleep and can obtain the required data from the patients (Ventouras, 2005).

The rest of the paper has been divided into following segments. Section 2 covers various intelligent computing techniques such as ANN, Fuzzy, GA, DM and integrated models like RBR–CBR, CBR-FL, RBR-ANN, ANN-FL,

ANN-BN, ANN-GA, ANN-DM, GA-FL. Section 3 deals with the obtained results from review. Section 4 and Section 5 covers the conclusion and summary. Section 6 covers all the references used for literature review.

2. INTELLIGENT COMPUTING TECHNIQUES 2.1 Knowledge based System (KBS) / Rule based Reasoning (RBR)

KBS is an artificial intelligent device to offer intelligent decisions with validation. In this knowledge acquisition and representation is being done by various rules, frames and scripts. Knowledge is represented by means of cases (CBR) and IF-THEN rules (RBR). The main components of RBR are rule base and inference engine. Rule base contains a list of rules considered as knowledge base. An inference engine infers information based on the interaction of input and the rule base. A match-resolve-act cycle is performed to execute the construction system program. The main advantage of RBR is the explanation of information in the form of rules, compress representation of rules and modularity. R&K's criteria are basically based on marking events such as sleep spindle, k-complex, rapid eye movements and slow delta waves rather than background signal activities. If no marking events are found at the sleep epoch, then the event-based classification and smoothing rules cause artificial neural network with low performance. Also rule-based expert system has two limitations: 1) all situations cannot be described by rules, 2) experiences acquired by trial and errors cannot be easily contained to the knowledge base without human effort and 3) Misconception to understand the rules. A neural network system comprised the shortcomings of a rule-based expert system. The limited reliability of automated sleep scoring was overcome by hybrid neural network and rule-based expert system with agreement rate of 85.9% (Park, 2000).

CBR is the process of solving new problems based on the solutions of similar past problems. CBR based systems attempts to produce a solution to a new problem by making use of the 4R's: Retrieve Reuse, Revise and Retain (Parka, 2000). CBR has Capability to convey specific knowledge and update the KBS whenever a new case arrives. It also helps to manage the unpredicted inputs. But there are also some limitations like as knowledge acquisition problems for unavailable or limited cases, Inference efficiency is not always good as desired, straight forward provision of explanation is missing. A few researchers have used the integrated approach of CBR and RBR for the implementation of systems to detect the sleep disorders as depicted in Table 6.

2.2 Artificial Neural Network (ANN)

ANN commonly used for pattern recognition and classification (Mendez, 2009) consists of a collection of perceptrons combined in layers by connectionist and data controlling approach in medical domain. It possesses an adaptive nature to change its structure during its learning phase, thus, used to solve real world complicated relationships. It pertains to problems in which the training data assimilate to noisy and complex sensor data as well as problems in which more symbolic illustrations are used. The BP algorithm is mainly used in ANN learning technique which has been used by many researchers in the detection and classification of SA/hypopnea events.

ANN has a few advantages over KBS in exhibiting a complementary approach to RBR in terms of knowledge illustration which requires a long time to construct such a system from rule based approach that extracts features from the original recordings like EEG, PSG and then make the rules according to human knowledge (Tian, 2005). ANN possesses a very attractive property for automated recognition of sleep EEG patterns which doesn't require any elaborate classification rules or complex domain knowledge (Ventouras, 2005). ANN agreed with manual scoring of 93.3% for all scored epochs to manual scoring of arousals, a time consuming process (Sinha, 2008). The semi-automatic arousal detection system was implemented by Sorensen using FFNN (Sorensena, 2011) to overcome the limitations of manual system. The abstraction and repetitiveness of the task also direct to inaccuracies and low inter-scorer agreement (Tian, 2005) wherein no knowledge of probability distribution is required. NN is skillful in estimating the posterior probabilities, providing the base for implementing classification rules (Marcos, 2008). Automated classifier (Flores, 2000) was proposed to detect the arousal, an inexpensive and less time consuming process as compared to other approaches. In automated sleep spindle detection studies, referenced to the visual detection results ranges from 70% to 99% and false positives from about 3% to 47% (Ventouras, 2005).

ANN is efficient in the classification of non-linear and non-periodic types of signals like as sleep EEG patterns. In another report, two types of ANN: MLP and LVQ are used to classify the sleep stages in infants. Automatic sleep stage scoring in human was tried by using multilayer FFNN with the recognition rate varying from 82.3-90% (Sinha, 2008). The advantage of BiosleepTM (oxford biosignals), which uses automated neural network technique for sleep staging for ease and speed of analysis. Not bounded to R&K criteria, it offers the benefits over manual assessment (Caffarel, 2006). ANN insensitive to the distributions of the data shows similar obtained results with raw or transformed data. ANN is deft in dealing with a non-Gaussian probability density function and extreme values. Transformations are ineffective on its ability to separate space into subspaces (Becq, 2005).

k-NN is a nonparametric technique for classification assumed to have no apriori parameterized knowledge about the probability structure of the data (Ventouras, 2012) and data samples are used directly where it provides best performance (Mendez, 2009) in homogeneous and transformed data (Becq, 2005) for enough number of cells with small size which are achieved by LDA as compared to ANN (Krajewski, 2007).

LVQ, a network with supervised training showed a better adaptation to the training set with an increasing number of neurons achieving maximum classification (Golz, 2001). Normal readings are separated entirely 100% from the apneic recordings from k-NN and NN supervised learning classifiers (Mendez, 2009). The advantages of RBF networks are: simplicity of the architecture, reduction in training time and the capability to deal with unseen data. RBF networks are successfully applied for fault detection, face recognition or medical diagnosis. RBF-FCM network provided the best classification accuracy with reduced network complexity whereas RBF-KM network provided the best classification performance as compared to RBF-OLS and RBF-FCM networks, as their performances were slightly lower than that of RBF-KM (Marcos, 2008).

Inspite of few benefits, ANN has certain drawbacks, like the organization of NN is ambiguous. A priori information/knowledge used for the initialization purpose cannot be treated for better initialization of network parameters and the reduction of learning time period. It is incorporated rule -based expert system which could not contain all the rules due to its very low agreement rate 55.1% compared with Rule-based expert system 83.0% (Park, 2000). The visual recognition and counting of spindles is a laborious and time-consuming task for wholenight sleep EEG recordings (Ventouras, 2012). There are problems associated with mimicking a human scorer in an automated sleep spindle detection system (Ventouras, 2005). Table-1 summarizes the various ANN approaches followed in the each specific selected paper that is used in the detection and classification of sleep stages and SA, OSA, sleep spindles. ANN based analysis system is not sufficiently accurate for sleep study analysis using the R&K classification system. BiosleepTM is limited to visual inspection between pseudo-R&K hypnogram and manual scorers (Caffarel, 2006). Leave one out or classical crossvalidation techniques cannot be applied when working with a large database. Classification errors reach approximately to 30% and do not improve when the number of data in the sets increases over 500 samples while using k-NN classifier and the Parzen estimator. MLP and k-NN provided good results as compared to another techniques but k-NN has a majority vote over the remaining classes. The selection of the best NN and the optimized structure of the layers are difficult and time consuming. k-NN requires space for a large amount of learning vectors in memory (Becq, 2005). The unsupervised SOM and GCS methods were understandably poorer (Golz, 2001). Among the apnea screening methods presented by de Chazal et al. is able to achieve 90% of correct classification on minute-by-minute basis. It extracts spectral features through fourier transform of both RR series and ECG morphological characteristics. In our opinion, its main weakness is in the high-dimension feature space, i.e., 88 different features. The choice of simple topology will result in a network that is incapable of learning a complex function, whereas complex topology leads to loss of generalization capability resulting in over fitting the training data. In complex structure, NN may have the ability to memorize the training set resulting in inaccurate predictions on future samples. So, early stopping technique is an alternative to it. It comprises a validation set to stop the training algorithm before the network starts learning noise in the data as part of the model leading to an estimate of the generalization error (Mendez, 2009). The best generalization performance is achieved by a network whose complexity is neither too small nor too large. RBF-KM network since increasing the size of the hidden layer did not substantially improve the accuracy. Moreover, training MLP networks with BP requires a larger number of user-dependant parameters to be a priori specified, such as learning rate, momentum or number of training epochs (Marcos, 2008). The further description can be found from the following Table 1.

SVM, due to its generalization ability is used for solving supervised classification, regression (Ronald Fisher, 1955) problems (Maali, 2012) and binary classification (Maali, 2012) tasks containing elements of non-parametric applied statistics which maximizes the margin between the training data and decision boundary, can cast as a quadratic optimization problem (Actr, 2004). It is a machine learning method proposed by Vapnik in 1995. The idea of SVM is to construct an optimized separating hyperplane. The optimization criterion of SVM is the width of the margin among the various classes i.e. the blank space around the decision boundary defined by the distance to the nearest training patterns (Maali, 2012).

SVMs are supervised learning models associated with all learning algorithms to analyze data and recognize patterns and map data into a high dimensional space to find out a separating hyper plane with maximal margin (Diw Berlin, 2008). The advantage of SVM is that it can solve the problem of non-linear classification and no need of velocity clamping for constriction model (Maali, 2012). The kernel function is used to solve the problem of inner product calculation in high dimension, hence, a good method for non-linear classification. The kernel function must be chosen to achieve the best classification accuracy for unknown samples (Maali, 2012). The cross-validation and independent test accuracies of apneic event detection were found to be 93.3% and 92.8%, respectively. For hypopnea event these two accuracies were 90.1% and 89.6% in which sensitivity was used to optimize the SVM parameters (Koley, 2012). After testing three different kernel functions such as: RBF, Sigmoid and polynomial, it

was found that polynomial kernel shows higher performance (Maali, 2012) than others. In comparison with

GAs, PSO have fewer complicated operations. So, few parameters can be coded dependent on stochastic processes to overcome it. SVM has ability to minimize both structural and empirical risk leading to better generalization for new data classification. Fast convergence is the one limitation of PSO (Maali, 2012). Self-advising SVM significantly provides better results than traditional SVM.

Self-advising SVM intended to deal with the ignoring of the knowledge extracted from the misclassified data. SVM is an approximate implementation method of structural risk minimization to attain low probability of generalization error. The classification performance of k-NN, PNN and linear discriminant classifiers on test data was lower than SVM. Three kernel functions are used such as: linear, polynomial and radial basis function, in which 100% accuracy is obtained using polynomial kernel function with four subsets of features similarly provided by linear functions with only two subsets of features. k-NN and PNN show poor classification performance i.e.,83% and 70%, respectively on the test data (Maali, 2012).

Table 1. Artificial Neural Network

Author	ANN System	ANN Model	Accuracy(Ac), Sensitivity(Se), Specificity(Sp)	Application
Sorensen et al.[15]	Semi-automated system	2-layer FFNN with hyperbolic tangent function for the hidden layer and a logistic sigmoid function for the output layer, leave one out method to validate the algorithm, Fourier Transform to assess the frequency content and PPV-88.8 %	Se-89.8%,	Detection of Arousals
Flores et al.[16]	Automated classifier	ANN, a single neuron with hyperbolic tangent activation function at output layer, LM algorithm to train network, novel pattern recognition algorithm, Fourier Transformation, south- well relaxation technique for adjustment of parameters, Multiple additive regression trees to benchmark ANN	Se: 96.1%(sleep), 95.25%(wake)	Classification of sleep versus wake stage
Ventouras et al.[17]	Automated system	Feedback MLPANN, Transient waveforms, specificity was calculated for the three groups: SS, Mild Cognitive Impairment (MCI) and Alzheimer disease(AD)	Se: 81.4%(SS), 62.2%(MCI), 83.3%(AD)	Detection of sleep spindles
Ventouras et al.[18]	Automated system	FFMLPANN of 3 layers with 64, 30 and 2 neurons in the input, hidden and output layer resp., feature extraction, time-domain representation of EEG signals, error back-propagation training algorithm, log sigmoid transfer function for neurons	Ac-92%	Detection of sleep spindles
Sinha[19]	Application	BPNN with 60 nodes used as input, 18 for hidden layer and one output layer. ANN was found effective to distinguish the EEG power spectra such as: 92% in sleep wake state, 85.5% in REM, 91% in awake stage	-	Detection of sleep- wake stages
Sinha[20]	Automated system	BPNN with 64 neurons as input, 14 hidden neurons and 3 neurons in output layer and average agreement rate was 95.35%	-	Detection of REM sleep, sleep spindles and wake states

Caffarel et al.[21]	Biosleep TM ANN	ANN, cohen's kappa coefficient for various combinations of sleep stages.	-	Detection of SS
Liu et al. [22]	ANN of ART2	ANN, ART2for the reduction of communication cost, fourier or wavelet transformation	Ac-91%	Detection OSA, narcolepsy
Norman et al.[23]	Automated system	BPNN, agreement of ANN was 89% in the training set and 70.6% in the test set	-	Detection of sleep disordered breathing
Krajewski et al.[24]	Approach	PCA for reducing dimensions, FFMLPNN with BP learning algorithm, one hidden layer and 5 nodes, LDA based classification models, two-fold cross-validation reduces effort	-	Detection of sleepiness
Marcos et al.[25]	Assistant tool	RBF network with one hidden, one output layer and used in function approximation, time series prediction, control and classification, k-means (KM) and fuzzy c-means (FCM) clustering algorithms allow to find centre positions in an unsupervised manner, whereas the OLS(orthogonal least squares) algorithm involves supervised training of RBF networks, ROC analysis to compare the classification ability, analyze the performance of all algorithms with RBF	RBF-KM Ac-86.1%, Se-89.4%, Sp-81.4%, RBF-FCM Ac-84.7%, Se-86.6%, Sp-81.9%, RBF-OLS Ac-85.5%, Se-89.8%, Sp-79.4%	Diagnosis of OSA
Golz et al. [26]	Vector based NN	LVQ, self-organizing feature map and growing cells structure for classification of spectral density	In general, Ac-80% but less on high dimensions data	Recognition of beginning microsleep
Mendez et al.[27]	Time varying auto regressive model	k-NN, FFBPNN with 1 neuron at output layer (linear neural function), 3 to 30 neurons at hidden layer (logarithmic Sigmoid neural function),sequential forward selection approach for the selection of best features, Wrapper approach through Leave One Out to compute classifiers performance, LM training procedure to train NN, early stopping for validation of NN	KNN: Ac-88%, Se-85%, Sp- 90%, NN: Ac-88%, Se-89%, Sp- 86%	Classification of OSA
Tian et al. [28]	TDNN	TDNN with 2 nodes at1 hidden layer, 3 nodes at 1 input layer and 2 nodes at 1 output layer, BP to train TDNN	Apnea: Se-90.7%, Sp-6.4%, Hypopnea: Se-80.8%, Sp:81.4%	Detection of SA
Almazaydeh et al.[75]	FFNN	Network was trained using BP learning algorithm and Spo2 signals are passed as inputs to ANN	Ac-93.3%, Se-87.5%, Sp-100%	Detection of OSA

Dolypona Das et al.[76]	ANN	Network was trained using BP learning algorithm Spo2 and ECG signals are used for the identification of OSA	Ac-97.3154%	Identification of OSA
Gabran et al.[77]	Real Time Automated System	Three classification methods are implemented: LVQ,PNN and FFNN, single hidden layer was selected in all techniques, 4 neurons in input layer and one neuron in output layer, tansig + purelin and logsig + purelin used as activation functions, leave-one out cross-validation technique	Ac: LVQ-70% PNN-79% FF-NN-85%	Classification of sleep
Tagluk et al.[50]	ANN	Bispectral: QPC, 2-D window function to reduce the variance of the bispectrum, hanning window for bisopectral calculation NN: one input layer, two hidden layers, and one output layer. output error was measured by means of SSE, multilayer NN for classification, cross validation technique	Ac-96.15%	Estimation of OSA
Becq et al. [54]	Analysis	Comparison of linear and quadratic classifiers, k-NN, Parzen kernels and neural networks in automatic sleep stage classification, MLPNN trained with 8 neurons in the input layer (hyperbolic tangent transfer function) 6 neurons in the hidden layer (linear transfer function and 6 in the output layer (logarithmic sigmoid transfer function) trained by with FFBP gradient algorithm	-	Scoring of SS
Lin et al.[74]	WT + ANN	Transformation: Wavelet transforms for the decomposition of EEG signals into delta, theta, alpha, and beta spectral components ANN: Wavelet coefficients for training purpose, 12 neurons in Input layer, 32 neurons in hidden layer and 3 neurons in output layer, sigmoid function for the restriction of output values	Se-69.64%, Sp-44.44%	Identification of SA
Huang et al.[83]	Automated system using ANN	Fully connected neural network, BP algorithm to minimize the root MSE, analyze the accuracy over three different approaches: 3-layer ANN(16-10-6), 4-layer ANN(16-10-8-6) and 2-layer ANN(16-6)	Ac: 3-layer-90.83%, 4-layer- 84.18% 2-layer-78.64%	Discrimination of SS
Roberts et al.[84]	Kohonen self- organising ANN	Kalman filter for the analysis of EEG signals and ANN for clustering in high dimensional space	Ac-80%	Indication of dynamics in sleep
Schaltenbrand et al.[85]	3-layer ANN	3-layer ANN was used in which 17 neurons in input layer, 10 in hidden and 6 in output layer	Ac-80.6%	Analysis of human sleep
Vijaylaxmi et al.[87]	WT - ANN	Features were extracted from EEG signals by discrete wavelet transform, FFMLP neural network(3-layer: 5-5-1) was used to classify the signals where supervised learning was provided by BP and average classification for 3 epochs 1000, 2000 and 3000 was 85%, 90% and 93%	Se-55.17%, Sp-80.25%	Classification of SS
Liu et al.[91]	ANN of ART2	Classification of excessive daytime sleepiness was done form normal subjects, Yoss pupil staging rule to check the wakefulness, Fourier or wavelet transforms	Ac: 91% (OSA) and 90% in narcolepsy subjects	Detection of OSA and narcolepsy
Emoto et al.[92]	ANN	3-layer perceptron NN was used as second order one step predictor, linear activation function used in output layer containing 1 neuron, hyperbolic tangent function used in hidden layer, independent component analysis on connection- weight-space approach	-	Snore Sound
Jane et al.[95]	FF multilayer NN	A detector is proposed including episodes such as: snores, sounds during inspiration and exhalation, speech and noise artifacts, 2 layer FF multilayer NN with 22 neurons in input	Se-76.1%, Sp-82.8%	Analysis of snoring signals from sleep

		and 50 in hidden layer and 2 neurons in output layer, PPV was 75.6%		
Gorur et al.[90]	STFT + ANN	Features were extracted by STFT and comparison was done by MLP and SVM for detection of the spindles in sleep.	Ac: MLP-88.7%, SVM-95.4%	Detection of sleep spindles
Berdiñas et al.[42]	ANN experts + SVM recursive feature elimination	Model was based on error correcting output code, inputs provided to the experts were the coefficients gained by a discrete wavelet decomposition, 10 different simulations and a multiple comparison procedure was used for model selection, 10 fold cross validation for error rate	Ac-90.27%±0.79	Classification of SA
Yildiz et al.[43]	LS-SVM + discrete WT + fast fourier transform	LS-SVM classifier was compared with RBF, linear and polynomial kernels using 10-fold cross-validation method, Performance was optimized using hill climbing feature selection algorithm	Linear: Ac-90%, Se-80%, Sp-86.7% Polynomial: Ac-85%, Se-60%, Sp-76.7% RBF: Ac-95%, Se-60%, Sp-83.3%	Recognition of OSA
Koley et al.[34]	Three binary SVM	leave-one-out cross-validation to evaluate the generalization ability, one-against-all strategy to train SVM, winner-takes-all for class label	-	Detection of SA and hypopnea events
Maali et al.[36]	Partially connected co-operative parallel PSO-SVM Algorithm	PSO: To tune SVM parameters and data reduction, to search space more efficiently parallel structure: master-slave(multi-swarms) with new cooperative strategy, swarms classified to several masters and one (or more) slave, avoiding fast convergence and local optimal results SVM: OSU-SVM package	Ac-87.93%	Detection of SA
Maali at al.[38]	Self-Advising(SA) SVM	PSO: To select best features subset and for tuning of SA-SVM parameters SA SVM: Used as fitness function, particle representation by two arrays, t-test for comparison between SVM and SA-SVM	Ac-83.93%	Classification of sleep apnea
Khandoker et al.[37]	SVM	To train the SVM classifier, features were extracted of heart rate variability (HRV) and ECG-derived respiration (EDR) signals, leave-one-out approach used for optimal SVM parameter set	-	Recognition of obstructive sleep apnoea
Almazaydeh et al.[73]	SVM	RR-interval based features of ECG signals are passed to SVM as inputs, liner kernel function for mapping of training data into kernel space, sequential minimal optimization process	Ac-96.5%, Se-92.9%, Sp-100%	Detection of obstructive sleep apnea
Gabran et al.[78]	Portable real time prototype based on	Three classification methods were analysed: LVQ, PNN and FFNN, single hidden layer was selected in all techniques, 4 neurons in input layer and one neuron in output layer, tansig +	Ac:	Automatic Diagnosis and

	SVM	purelin and logsig + purelin used as activation functions, leave- one out cross validation technique	LVQ-70% PNN-79% FF-NN-85% SVM-96%	Detection of Narcolepsy Episodes
Ieong et al.[81]	SVM	SVM was applied to check the snoring level (heavy and light) from heart rate variability features, PCA was used to increase the classification accuracy, Snoring Density was calculated, Done comparison between SVM and statistical analysis in which SVM provides better results as compared to statistical methods	Ac-75.82%	Snoring Classifier
Estevez et al.[52]	SVM-ANN	Classifiers were analyzed for the learning process	Ac-0.92%	Detection of EEG arousals
Liao et al.[106]	SVM	multivariate adaptive regression splines (MARS) and support vector machine was used for classification using acoustic features for categorization of simple snores and patients suffered from OSA	-	Diagnosis of snoring sound

Notes:- ANN: Artificial Neural Network, ART2: Adaptive resonance theory, BP: Back Propagation, ECG: electroencephalography, EEG: Electroencephalograms, EOG: Electrooculograms, EMG: Electromyograms, FF: Feed Forward, k-NN: k-Nearest Neighbors, LM: Levenberg-Marquardt, LS-SVM: Least squares support vector machine, LVQ: Learning Vector Quantization, MLP: Multilayer Perceptron, MSE: Mean Squared Error, OSA: Obstructive Sleep Apnea, PCA: Principal Component Analysis, PNN: Probabilistic Neural Network, PPV: Positive Predictive Value, PSO: Particle Swarm Optimization, QPC: Quadratic Phase Coupling, RBF: Radial Basis Function, ROC: Receiver Operating Characteristic, SVM: Support Vector Machine, SSE: Sum Square Error, STFT: Short Time Fourier Transform, TDN: Time Delayed Neural Network, WT: Wavelet Transform

2.3 Fuzzy Logic/Reasoning (FL)

Fuzzy set theory plays a vital role in dealing with complexities when drafting decisions in medical domain (Liang, 2011). FL is a type of probabilistic logic which deals with reasoning that is approximate rather than constant and accurate. FL variables may have a truth value that ranges in degree between 0 and 1. It has been expanded to manage the theory of partial truth, where the truth value can have range between completely true and completely false. In linguistic variables these degrees may be handled by specific functions (Riko safaric and

Andreja Rojko (2006). Fuzzy rule based system achieved good results for some samples, but it still needs to improve its performance in all samples of recording data (Maali, 2012). It had overcome the limitations of epoch-based sleep staging by obtaining a more continuous evolution of the sleep of the patient. Avoidance of binary decisions provides soft transitions and enables concurrent characterization of the different states of sleep. The use of Mamdani-like fuzzy rules enables knowledge to be implemented in form of linguistic rules, close to human language, which facilitates knowledge acquisition, understandability and allows explanatory capabilities (Estevez, 2012). The limitation of R&K rules was the unnatural assignment of discrete stages aside doing it in a continuous manner (Estevez, 2009). A receiver operating curve (ROC) index of 0.88 was obtained for the classification of events as apneas or hypopneas (Estevez, 2009). The manual sleep classification of the patients was made by different experts with an inter-rater reliability of 71%. The clusterization process has higher results than those from the process of discretization (Pinero, 2004) as shown in following Table-2. It has many advantages like as easy reformation of rule base or fuzzy sets, ease of understanding due to the presence of output in linguistic form, easy to design due to less cost, provisions to allow conflicting inputs, arrive at steady state in a lesser time interval. On the other side, fuzzy has also limitation like as difficult construction of a model from a fuzzy system.

Table	2.	Fuzzy	Reasoni	ng
-------	----	-------	---------	----

Author	Fuzzy Reasoning	Accuracy(Ac), Sensitivity(Se), Specificity(Sp)	Application Domain
Estevez et al.[30]	111 fuzzy rules, trapezoidal fuzzy sets, defuzzification is done by the center-of-gravity method, optimal parameterization of the fuzzy sets is done by GA, Cohen's kappa index values	-	Analysis of the sleep macrostructure
Estevez et al.[31]	Two layer architecture, five parameters are passed as input, trapezoidal fuzzy sets, defuzzification is done by the center-of-gravity method	Ac-95.6%, Se-95.5%, Sp-95.1%	Evaluation of the awake sleep state

Estevez et al.[32]	Fuzzy rules, trapezoidal fuzzy sets, input and output variables are fall reduction and event desaturation, Defuzzification is done by centroid method	Se-87%, Sp-89%	Detection of apneic events in SA/ hypopnea
Pinero et al.[33]	Five layer architecture, GENRUL5 and MLRUL algorithms are used for fuzzy rules, fuzzy pack, hard rule system, CBR used in inference process, fuzzy clustering	-	Classification of SS
Causa et al.[29]	Nonlinear algorithms + empirical-mode decomposition, Hilbert-Huang transform	Se-88.2%, Sp-89.7%	Novel tool for automated REM's
Liang et al.[82]	14 fuzzy rules are included in inference system in which 11 are referred to relationship between epochs, nine features passes as input to fuzzy system	Ac-87.05%, Se- > 87%	Detection of SS
Yadollahi et al.[86]	Energy of breathing sound signals was decomposed into two segments i.e., sound and silent segments and passed as an input to the fuzzy system in which sigmoid function was used for fuzzification, correlation between proposed system and PSG was 96%	Se and sp >90%	Detection of apnea and hypopnea events

2.4 Genetic Algorithm (GA)

Genetic Algorithm is a search heuristic optimization technique, which mimics the procedure of natural development. They have been exposed to find out the optimal solution for several difficult problems as an effective and suspected tool based on principles of evolutionary strategy (Maali, 2012). It is also possible to transfer the GA to existing simulation models. A classic genetic algorithm needs two terms: an inherited demonstration of the solution domain and a robustness function to assess the solution domain. They are valuable aspirant for parallelization (Maali, 2012) as depicted in following Table 3. It also provides the automatic scores based on sleep stages (Herscovici, 2007). The main advantage of GA is that no condition to have knowledge of mathematics for understanding. It is best to use in case of huge and complex search domain. In these cases local minima of price function can be easily obtained from gradient optimization method (Herscovici, 2007). There are also some limitations of GA such as no constant optimization response time, no assurance for global optimum result, limited control over applications of GA in real time and perfect optimization problems cannot be solved by GA (Riko safaric and Andreja Rojko (2006).

Table 3.	Genetic	Algorithm
----------	---------	-----------

Author	Remarks	Accuracy(Ac), Sensitivity(Se), Specificity(Sp)	Application Domain
Herscovici et al.[39]	GA was used to maximize the best score, iterative simulation was performed till the achievement of best genomes. Agreement rate- 89%, area under curve is 88%	Se-78%, Sp-92%	Detection of REM sleep
Sun et al.[105]	3 Questionnaires were involved to find out the OSA that was compared with logistic regression model and it was found that sensitivity and specificity of the logistic regression model were less than GA i.e.,55.6% and 57.9% respectively.	Se-81.8%-88.0%, Sp-95%-97%	Detection of OSA

2.5 Data Mining (DM)

Data mining is the process used in knowledge discovery process for pulling out previously unidentified, unforeseen relationships and layouts (Ho Viet Lam and Nguyen Thi My Ding, 2013). DM provides offers to distinguish highly productive chunk of information covered in the expensive and huge databases. The statistics provided by DM packages are: Decision tree (C4.5, CART, CHAID), Rule induction (AQ, CN2, Recon), nearest neighbors (CBR), Clustering methods (data segmentation), Association rules. Tree-like structures are used to improve the accuracy over a single tree structure (Khasawneh, 2012). If the variation of neurons is increased in the hidden layer, then mean of accuracy is also increased and standard deviation decreased. The good performance is achieved by using the trainlm (LMBP) and traingdx (Gradient descent with momentum) and adaptive learning rate BP training functions to train the ANN, but the trainlm produces better results in training and testing as compared to traingdx (Ebrahimi, 2008). The prevalence of Insomnia occurred frequently in OSAS patients and was systematically related with poor sleep quality but had no affect on long term complications and according to (Nguyen, 2010), 50% patients suffered from insomnia with range from moderate to severe. The aggrement rate was extended by extracting features from EOG and EMG signals from 71% to 80% (Zoubek, 2007). An automated detection system was proposed by (Shmie, 2009) to avoid the time consuming manual arousals process.

Table 4. Data Mining

Author	Technique Used	Remarks	Accuracy(Ac), Sensitivity(Se), Specificity(Sp)	Application Domain
Ebrahimi et al.[53]	Automated system following tree structure	MLPANN with 12 inputs, one hidden layer with 8 neurons (asymmetric sigmoid function) and one output layer with 4 nodes with traingdx training algorithm, Boot strap technique, LM as training function, Comparison of linear and quadratic classifiers, k-NN, Parzen kernels and neural networks in automatic sleep stage classification	Ac-93.0 ±4.0%, Se-84.2+3.9%, Sp-94.4±4.5%	Classification of SS
Khasawneh et al.[40]	Multiple Decision Tree	Two algorithms are used as: pre–pruning + true positive rate and the other is based on maximum probability voting	7% more accuracy over a single model	Classification of SS
Nguyen et al.[100]	Clementine Software (SPSS)	Overweighting technique to rebalance the sample size, decision tree segmentation, Z-test for comparison, chi- square test, pearson correlation test, insomnia severity index was 15.	-	Insomnia symptoms and CPAP compliance in OSAS patients
Shmie et al.[102]	DM	Meta rules are used to detect the arousals, PPV=76.5%	Se-75.2%	Detection of sleep arousals

2.6 Bayesian Network (BN)

The bayesian networks provide a model for undecided reasoning, capable for handling the diagnostic problem. BNs are strong and healthy formalism that allow reasoning under uncertainty, proposing a graphical representation of statistical dependencies between domain variables (Milho, 2000).

A BN is a directed acyclic graph (DAG) i.e., the combination of various nodes which signify random variables are connected by various edges that characterize the conditional probabilistic dependency between all vertices Scott McCloskey (1999-2000). The division of every node is dependent on its parent nodes and we can say that any node Y is conditionally independent of X if there is no straight path from X to Y. So, BN is constructed to signify not only correlation but causality. It facilitates visualization of the strongest links (Winrow, 2009). A few researchers have adopted BN for analysis and diagnosis of sleep apnea as described in the following Table 5. DiBa algorithm has less complexity, soft, high resolution output,

online implementation and displays detection performance used for the detection of sleep spindles. This algorithm depends on analyzing the time-varying posterior probability to detect spindle events. This allowed us to measure the method's receiver operating characteristic (ROC) curve by varying the likelihood threshold through all possible values. The ROC value achieved for this method is 0.98 and it is better as compared to other methods with respect to flexibility, transparency and scalability (Babadi, 2012). BN can manage partial/incomplete data sets and there is no need of preprocessing. On the other side, BN can have lagging of good classifier in the presence of wrong preceding knowledge, required a large amount of efforts to construct a network and difficult to manage constant features. Moreover, hypothesis space is also predetermined. As compared to linear classifier, Naive bayes provide better results in the identification of sleep stages with accuracy of 70% (Sloboda, 2011).

Fable	5.	Bayesian	Network
		,	

Author	Technique Used	Remarks	Application Domain
Winrow CJ et al.[11]	BN	To determine the sub-networks of linkage and loci that was not measurable in smaller data sets	Analysis of sleep-wake traits
Milho et al.[12]	BN	Statistical inference algorithm, java interface module, three nodes D,S,TR, handling of typical queries such as diagnostic, causal, inter-causal and mixed inferences	Sleep disorders diagnostic system
Saat et al.[13]	Empirical Bayes- Poisson-Gamma Model	The assessment of parameter was done by three methods: moment (mom), maximum likelihood (mLE) and alternate method (mLM)	Occurrences of SA and REM sleep stages

Babadi et al.[14]	Data Driven Bayesian Algorithm	Karhunen–Loeve transformation and Bayesian hypothesis, extraction of data-driven features by DiBa algorithm	Detection of sleep spindle
Sloboda et al.[88]	Linear and naïve bayes classifiers	Extraction of respiratory signals were done by band pass filter,10 fold cross validation technique for checking errors	Identification of SS

Notes:- BN: Bayesian Network, REM: Rapid Eye Movement, SA: Sleep Apnea, SS: Sleep Stages

2.7 Integrated Computing Techniques

A standalone KBS system have their own benefits and limitations like as inference problem, knowledge acquisition problem etc. as already covered in Section 3.The limitations of the standalone KBS are eliminated by integrating the KBS and ICT approaches.

In this section, we have covered the advantages incurred from integrated model along with their applications in the diagnosis and treatment of the sleep disorders. The combination of CBR-RBR leads to the ease of knowledge acquisition, improvement achieved in terms of accuracy, performance and efficiency. The main advantages prevailed from the integrated techniques are such as high performance on sleep disordered recordings, the justification and the learning capacity. The area of sleep stage scoring is thought to require hybrid reasoning because human experts use both rule-based knowledge and their experiences (Parka, 2000). To present the advantages of CBR in the diagnosis and treatment of OSA, the authors (Kwiatkowska, 2004) developed a small prototype system called Somnus which provides case storage and retrieval in the sleep disorders. To deal with the diversity and complexity of data, the authors had used a framework combining the fuzzy logic approach for modeling of the case features and the semiotic approach for the modeling of their measures. The user interface built in the first prototype of the Somnus system is restricted to SQL like statements providing a quick access to data but now the system is limited to handle a small group of users familiar with the database schema. The use of semio-fuzzy approach provides us a uniform representation for quantitative, qualitative features, subjective and objective measures. Even the limited cases of CBR in Somnus can be helpful for the various health providers in the field of training (Kwiatkowska, 2004). Most of the computational system are proposed in the field of sleep are only dealing with the detection system. The first application of neural network in the era of sleep is reported in (Clabian, 1996 and Clabian, 1997). BP method is considered in this case but the classification rate didn't exceed 60%. Romero (2005) had tried three different supervised learning methods such as scaled conjugated gradient (SCG), regularized scaled conjugated gradient (RSCG) and a Bayesian approach. The SCG was selected as learning algorithm for the network due to its fast convergence speed and low memory requirements and mean squared error (MSE) as cost function. But the second approach (RSCG) is opposite to SCG in which MSE with a regularization term called weight decay is used as cost function. In which only the best results was achieved by using the bayesian framework and a regularized cross-entropy function by reducing the error function (Romero, 2005). A discrete wavelet transformation was used as a pre-processing phase to reduce and fix the number of inputs of the classifier. The selected Bayesian network has the advantage that all the learning parameters are auto-adaptive and no externalhuman action is needed. In previous research, not much emphasis had been located on detection and classification

of nonlinear properties in the EEG signals. The major cause can be the complex mathematics and we need expertise to interpret it. But in (Estevez, 2009), author has proposed the integrated combination of bispectral analysis with ANN. In contrast to power spectrum, bispectrum reveals a non-Gaussian and nonlinear information which allows the detection of nonlinear characteristics and characterization of nonlinear mechanisms. Quadratic phase coupling (QPC) is a unique property of bispectrum that can be used to detect, quantify and differentiate the normal from OSAS patients. Training is controlled by cross validation technique (Estevez, 2009). The chromosomes were designed with the variable length structure and the fitness function was used to find the optimal input features and recognition of network specifications. The integrated system based on the feed-forward neural network had evolved by the genetic algorithms to find the optimal structure and input features (Kim, 2000). The large amount of data, complexity of the classification analysis and the variability among human experts are reasons to develop an automated sleep classification system. Neuro- fuzzy was chosen for the construction of rules, for the classification process and to find the parameters that define the degree of presence or absence for a pattern. The results obtained from a multi-layer perceptron neural network is 87.3% and from an expert's rules for sleep classification is 86.7% while the achievement of results in Neuro-Fuzzy classifier is 88.2%. A long time is needed to construct system from rule based approach for feature extraction from the original recordings. According to the R&K criteria, each sleepwaking state must last at least one minute. It is impossible to transit directly from awake to sleep 4 for a healthy person. Using SOFM's self-organizing abilities it is possible to train the ANN with a large number of EEGs (Tian, 2005). Sleep staging algorithms can be generally organized into 2 main categories: methods based on rule based reasoning and based on numerical classification (Koton, 1988). Rule Based Reasoning methods try to cover the R&K manual in a rule based form (Montani, 2003, Zurada, 1992 and Bernstein, 1995). The main hypothesis behind these methods is that many of the rules lead towards structured reasoning. Numerical methods govern most of the sleep studies with spectral analysis having the longest tradition of all EEG analysis techniques (Vinterbo, 2000 and Winrow, 2009). Also it is limited in case of accuracy and applicability. Neural networks were used to improve the automatic classification of sleep stages based on prior feature detection. The adaptive sleep staging system provides better reults than conventional rule based or numerical methods (Hassaan, 2008). An automatic diagnosis system must, therefore, be developed to reduce doctor's labor and realize quantitative diagnosis of sleep EEG. In comparison of detection performance with MLP, RB-SVM has been found the best with an average false detection rate of 4.0%. RB-SVM which is very useful due to its generalization ability in two stage classification procedure achieves a significant improvement in terms of sensitivity and false detection rate. Radial based SVM gives better results than the back propagation multilayer

perceptron (Acir, 2004). The agreement percentage was different in relation to total number of events for 406 and 553 that was 99% and 91% respectively (Maali, 2012). The results obtained in case of individual sleep stage classifier was increased from 76% to 97% (Mota, 1999). The results produced by genetic fuzzy classifier look

similar to visual sleep staging around 84.6% in spotting of wakefulness, shallow sleep, deep sleep and rapid eye movement stages (Han, 2010). The following Table 6 depicts the whole information related to various integrated techniques.

Table 6.	Integrated	Computing	Techniques
----------	------------	-----------	------------

Author Name	Integrated Components	CBR,RBR,ANN,GA,BN,FR	Accuracy(Ac), Sensitivity(Se), Specificity(Sp)	Application
Parka et al.[44]	CBR-RBR	RBSU:IF-THEN rules CBSU: case representation via attribute value, case retrieval by nearest neighborhood algorithm. When RBSU fails then CBSU is used. Enhanced agreement rate by 5.6% from the previous 87.5%	-	Automation of sleep stage scoring
Prentzas et al.[45]	CBR-RBR	The hybrid approach provides a clear understanding of the structure that makes system evaluation much easier, indentifies the unexplored cases, two main categories are illustrated in the schemes are standalone and coupling	-	Classification of integrating CBR and RBR approaches
Kwiatkowska et al. [47]	CBR-FL	CBR: case representation via multidimensional model and case retrieval via crisp value Fuzzy: linguistic variables for the representation of features and fuzzy membership functions for specification	-	Diagnosis of OSA
Romero et al. [46]	FFNN-BN	FFNN: 1 hidden layer BN: Soft max activation function to interpret the outputs near to probability, cross entropy function, cost function, Quasi-Newton optimization algorithm for optimum weight vector, 10-fold cross validation	Ac-83:78 ± 1:90%	Classification of SA
Heiss et al.[55]	Neuro-Fuzzy	NN: MLPNN with 5 input nodes, a hidden layer with 10 nodes and 5 output nodes Fuzzy: 32 rules, sigmoidal fuzzification function, two fuzzy concepts: squared error as objective function, pruning algorithm for reduction of rules, state duration algorithm	Ac-88.2%	Classification of SS
Park et al.[57]	Hybrid NN- Rule-based ES	NN: multilayer FFNN with two hidden layers, Error BP Learning method, LM weight learning algorithm Rule-Based ES: single epoch rule, three-minute smoothing rule	Ac-85.9%	Detection of SS scoring
Tian et al.[58]	NN-Fuzzy-RBR	Fuzzy: Triangular membership function, Fuzzy rule based classifier Characteristics of waves: K-complex, spindle detected by Object Oriented method, Kohonen algorithm used for calibration of self-organizing feature map	Ac-85.3%	Detection of SS
Hassaan et al.[59]	RBR-NN	RBR: classification rules based on R & K and AASM manuals, RBR engine to reduce false positive rate NN: FF Multilayer ANN, BP, Tansig, Purelin function	Se-29.53%, sp- 94.69%	SS scoring

Acir et al.[51]	BPMLP-(RB- SVM)	MLP trained by BP algorithm, bipolar sigmoidal function Novel ANN method, Adaptive autoregressive model for feature extraction	Se-94.6%	Automatic recognition of sleep spindles
Huang et al.[79]	GA-BPN	 GA: To forecast the relationship between insane proportions of health examination and lifestyles, linear order crossover was adopted of rate 0.8, swap mutation procedure BPN: To predict morbidity, sigmoid as an activation function, Total neurons of the input layer, hidden layer and the output layer are 2, 3 and 1 respectively 3-D bezier Surface is used to reflect the association between lifestyle and disease 	-	Application in health management system
Bellos et al.[80]	rule-based system-SVM Random Forests, ANN(MLP),Dec ision Trees, Naive bayes	DSS was analyzed on 3 databases. By applying Ist database, It was found that naïve bayes had achieved high accuracy by using Correlation-based Feature Subset Selection. From second database, Random Forest, MLP and SVM have high classification accuracy i.e., 98.75 % but by applying third database naïve bayes, MLP and SVM, all have achieved accuracy 73.08%.	MLP and SVM achieved Ac of >98%	Chronic disease management
Baile et al.[89]	Combination of all classifiers	10 machine learning algorithms (AdaBoost with Decision Stump, Bagging with REPTree, and either k- NN or Decision Table) were analyzed for classification and it was found that Spo2 features provide better result as compared to ECG features.	Ac-82%, Se-82%, Sp-82%	Detection of real time SA and hypopnea
Maali et al.[35]	Genetic-SVM (Novel system)	GA: Chromosomes are array with lengths equal to number of features, One point crossover and 10 random cells for mutation operations, fitness function SVM: OSU-SVM package	Ac-89.26%, Se-0.84%, Sp-0.88%	Detection of SA
Maali et al.[41]	GA-Fuzzy	Fuzzy: Mamdani fuzzy rule based system, Fuzzy rules are produced by GA GA: Three sets of chromosomes in the form of array, two cells for cross over and one for mutation	-	Detection of SA/ hypopnea
Zoubek et al.[101]	DM-MLPNN	 Two bayes rule based classifier, sequential method for feature selection 3 layers NN, hyperbolic tangent function, hidden and output layer contains 6 and 5 neurons respectively, logarithmic sigmoid function, FFBP gradient algorithm to train the net 	-	Classification of sleep/wake stage
Mota et al.[103]	GA-Fuzzy	Fuzzy: 1 input variable per feature, 3 membership function for output variable, If-Then rules with AND operator, 10 rules in rule set GA: Random initialization, roulette wheel for reproduction, uniform crossover operator, mutation operator	-	Classifier of sleep staging
Jo et al.[104]	GA-Fuzzy Classifier	Fuzzy: 5 input and 1 output set GA: Chromosome was conveyed as 171 bits, membership function as 75 bits and fuzzy rules selection as 96 bits, roulette wheel selection method, fitness function will be mean error rate, 0.8 cross over rate.	Ac-81.7% - 86.6%, Se-83.9%- 86.4%, Sp-79.6% - 87.3%	Identification of SS

3. RESULTS AND DISCUSSION

In this study, an attempt was made to study the various techniques adopted for the sleep disorders such as CBR, RBR, ANN, Fuzzy, GA, DM, BN and seven integrated methods such as CBR-RBR, CBR-FL, RBR-ANN, ANN-SVM, ANN-FL, ANN-BN, ANN-GA, GA-FL. Table-7 depicts the total number of cases in which various approaches are used for the detection, classification and diagnosis of sleep disorders. Table-9 contains the year wise information to adapt the different techniques for detection of sleep disorders. It is examined form Table 7 that in standalone category, out of 79 cases, only 5 cases of

BN, 7 cases of FUZZY, 2 cases of GA, and major cases of ANN (41) are used in the era of sleep disorders detection, classification and diagnosis. A few cases of integrated techniques were observed in this era such as: CBR-RBR(2), CBR-FL(1), RBR-ANN(2), ANN-FL(2), ANN-BN(5), ANN-GA(4), ANN-DM(1), GA-FL(3). It is noticed that among 79 cases referring to the practice of above techniques in sleep disorders, most of the approaches, i.e., 59 are utilized by only single methodology while rest of 20 cases are positioned by integrated approach.

Table 7. Comparison of numerical measurement of different computing tech	iniques
--	---------

	Intelligent Computing Techniques												
		St	andalone	e			Integrated Techniques						
Application	BN	ANN	GA	FUZZY	DM	CBR- RBR	CBR- FL	RBR- ANN	GA- FL	ANN- FL	ANN- BN	ANN- GA	ANN- DM
Detection	2	22	2	4	1			1	2	1	2	2	
Classification	2	14		2	3	2		1	1	1	2	1	1
Diagnosis	1	5		1			1				1	1	
Aggregate	5	41	2	7	4	2	1	2	3	2	5	4	1
%Used	8	69	3	12	7	10	5	10	15	10	25	20	5





We have also found the relative use of each technique with respect to total cases using single methodology which is represented by (m, p%), Where m is number of cases using a particular single methodology and p is percentage ratio of m (41) to total cases using single methodology (41+5+2+7+4=59) as presented in row7 (Aggregate) of Table 7. So, the relative use of ANN methodology is (69%), GA (3%), BN (8%), DM (7%), FL (12%) as

depicted in Table 7. For integrated techniques, similar calculations are used and represented by (i, q%), Where i represents total cases using a particular integrated technique and q is percentage ratio of i to total number of integrated techniques used. Hence, the relative use of CBR-RBR is (10%), CBR-FL (5%), RBR-ANN (10%), ANN-FL (10%), ANN-BN (25%), ANN-GA (20%), ANN-DM (5%), GA-FL (15%) as presented in Table 7.

	Intelligent Computing Techniques												
Application		St	andalone			Integrated Techniques							
	BN	ANN	GA	FUZZY	DM	CBR- RBR	CBR- FL	RBR- ANN	GA- FL	ANN- FL	ANN- BN	ANN- GA	ANN- DM
Detection	40	54	100	57	25	0	0	50	67	50	40	50	0
Classification	40	34	0	29	75	100	0	50	33	50	40	25	100
Diagnosis	20	13	0	14	0	0	100	0	0	0	20	25	0

Table 8. Comparison of numerical measurement in percentage of different computing techniques



Fig 2: Comparison of computing methods with respect to their percentage use

In this study, we have also reckoned the relative use of diagnosis, classification and detection in each standalone methodology and integrated approaches which are presented in Table-8. The findings in Table-8 are the ratio of the every component of first column to the last component of column like 40% is the ratio of first component of first column i.e., 2 is to 5. Similarly, second and third components are 2 is to 5 and 1 is to 5. Similarly, all components in the every column of Table 8 are

obtained according to components as per computation illustrated above.

It is noticed from figure s1 that mostly applications in sleep deploy ANN methodology and in few cases Fuzzy and BN have considered. Most of the integrated computing techniques deployed using ANN-BN and ANN-GA integration while other methodologies such as RBR-ANN, CBR-FL, ANN-FL, ANN-GA, ANN-DM, GA-FL and CBR-RBR are used in very few applications.

Table 9. Comparison of yearly numerical measurement	t of different	computing techniques
---	----------------	----------------------

	Intelligent Computing Techniques													
Year			Integrated Techniques											
	BN	ANN	GA	FUZZY	DM	CBR- RBR	CBR- FL	RBR- ANN	GA- FL	ANN- FL	ANN- BN	ANN- GA	ANN- DM	total
1990-1995		2							1					3
1996-2000	2	1				1		1						5
2001-2005		9		1			1			2	1			14
2006-2010		16	1	4	3	1		1	1				1	28
2011-2013	3	12	1	2	1				1		1	2	1	24



Fig 3: Comparison of computing methods with respect to their usage yearly

It is found from figure 2 that sleep applications deployed using GA are for detection while least for diagnosis in ANN methodology and high classification achieved using DM as they have been implemented using decision tree and association rules. In case of integrated techniques, highest classification is achieved by ANN, diagnosis by CBR-FL and detection by GA-FL.

It is observed from figure 3 that most of the applications in sleep disorders deployed using ANN in the period of 2006-2013 i.e., 16 and 12 in number respectively. Most of the implementations are done during 2006-2010 i.e., 28 and in 2011-2013, the number was 24.

4. CONCLUSION

The paper intentions are delivering a comprehensive scene in the development and deployment of various ICM in the era of sleep disorders. We have done a survey of the papers in this domain from 1991 to 2013s, covering 68 papers in the areas of sleep. It is examined that ANN approaches in particular, are widely used in the domain of sleep disorders as compared to other single technique where less reasoning is required. Integrated techniques are also used in the detection, classification of sleep disorders but as compared to single technique, their number is less. Out of 8 integrated approaches, ANN-BN and ANN-GA have been widely used percentage wise than other integrated approaches i.e. 25% and 20% respectively. From the previous past years, the trend of using fuzzy reasoning and data mining is also increasing. BN, ANN and Fuzzy are all mainly used for detection, classification and diagnosis purpose, GA is only used for detection while DM is used for both detection and classification. Our study would be helpful for the novices that may emerge their research in other medical domains.

5. SUMMARY

In this paper, we have done a review of the various intelligent computing methods and their combinations for the detection, classification and diagnosis of sleep disorders such as sleep apnea, insomnia, parasomnia and snoring. ICM comprises KBS, ANN, GA, DM, BN and FL. The integrated techniques are RBR–CBR, CBR-FL, RBR-ANN, ANN–GA, ANN-FL, ANN-GA, ANN-DM and GA-FL. The surveillance is done to explain the supreme and percentage use of different intelligent computing methods and their intra or inter combination

techniques. It is noticed that mostly ANN, GA and their integrated methods are used in the prognosis of sleep disorders.

6. REFERENCES

- Aamodt A., Plaza E. (1994) 'Case-Based Reasoning: Foundational Issues, Methodological Variations, and System Approaches' AI Communications. IOS Press, vol. 7, Issue 1, pp. 39-59.
- [2] Acır Nurettin, Guzelis Cuneyt (2004) 'Automatic recognition of sleep spindles in EEG by using artificial neural networks' Elsevier: Expert Systems with Applications, vol. 27, Issue 3, pp. 451–458.
- [3] Agarwal R. et al. (Aug. 2005) 'Detection of rapid-eye movements in sleep studies' IEEE:Transactions on Biomedical Engineering, vol.52, Issue 8, pp.1390-1396.
- [4] Almazaydeh et al. (2012) 'A Neural Network System for Detection of Obstructive Sleep Apnea through SpO2 Signal Features' in: Proceedings of International Journal of Advanced Computer Science and Applications, vol.3, Issue 5.
- [5] Almazaydeh Laiali et al. (March 29, 2012) 'Obstructive Sleep Apnea Detection Using SVM-Based Classification of ECG Signal Features' in: Proceedings of 34th Annual International IEEE, EMBS Conference.
- [6] Babadi Behtash et al. (Februarry 2012) 'DiBa- a datadriven bayesian algorithm for sleep spindle detection' IEEE: Tansactions On Biomedical Engineering, vol. 59, Issue 2, pp. 483-493.
- [7] Becq G. et al. (2005) 'Comparison Between Five Classifiers for Automatic Scoring of Human Sleep Recording' Studies in Computational Intelligence (SCI), vol. 4, pp. 113-127.
- [8] Bellos Christos et al. (August 30 September 3, 2011) 'Heterogeneous Data Fusion and Intelligent Techniques Embedded in a Mobile Application for Real-Time Chronic Disease Management' in: Proceedings of 33rd Annual International Conference of the IEEE, EMBS.

- [9] Bernstein A.D.et al. (1995) 'Diagnosis and management of pacemaker-related problems using and interactive expert system' in: IEEE 17th Annual Conference on Engineering in Medicine and Biology Society, vol. 1, pp. 701–702.
- [10] Bertha Guijarro-Berdinas et al. (2012) 'A mixture of experts for classifying sleep apneas' Elsevier: Expert Systems with Applications, vol. 39, pp. 7084–7092.
- [11] Caffarel J et al. (2006) 'Comparison of manual sleep staging with automated neural network-based analysis in clinical practice' Springer: Medical and Biological, vol. 44, pp. 105-110.
- [12] Canisius Sebastian et al. (August 20-24, 2008) 'Detection of Sleep Disordered Breathing by automated ECG analysis' in: Proceedings of IEEE EMB 30th Annual International Vancouver, British Columbia, Canada..
- [13] Causa L. et al. (Sept. 2010) 'Automated Sleep-Spindle Detection in Healthy Children Polysomnograms' IEEE Transactions on Biomedical Engineering, vol. 57, Issue 9, pp.2135-2146.
- [14] Clabian M, Nussbaum C, Pfutzner H. (1996) 'Artificial neural networks for apnea detection' Proc EANN, pp.601-608.
- [15] Clabian M, Pfiitzner H. (1997) 'Determination of decisive inputs of a neural network for sleep apnea classification' Proc EANN, pp. 171-178.
- [16] Das Dolypona, Vedanarayanan (February 2013)'Identification of Obstructive Sleep Apnea Through Spo2 and ECG Signal Features By Using an Efficient Neural Network System' International Journal of Advanced Research in Computer Science and Electronics Engineering (IJARCSEE) vol. 2, Issue 2, pp. 145-153.
- [17] Diw Berlin 'Support Vector Machine' available at http://www.diw.de/documents/publikationen/73/diw_ 01.c.88369.de/dp811.pdf (accessed on 7 March 2013).
- [18] Donald L. Bliwise et al. (1999) 'Correlates of the "don't know" response to questions about snoring' am j respir crit care med., vol. 160, pp. 1812–1815.
- [19] Dragulescu D., Albu A. (2007) 'Expert system for medical predictions' in: 4th International Symposium on Applied Computational Intelligence and Informatics, pp. 13–18.
- [20] Ebrahimi et al. (20-25 Aug. 2008) 'Automatic sleep stage classification based on EEG signals by using neural networks and wavelet packet coefficients' in: Proceedings of 30th Annual International Conference of the IEEE on Engineering in Medicine and Biology Society, EMBS, pp.1151-1154.
- [21] Emoto T. et al. (August 23-26,2007) 'Feature Extraction for Snore Sound via Neural Network Processing' in: Proceedings of the 29th Annual International Conference of the IEEE EMBS.
- [22] Espie Colin A., Brooks D.Neil (June 1989) 'An evaluation of tailored psychological treatment of insomnia' Elsevier: Journal of Behavior Therapy and

Experimental Psychiatry, vol. 20, Issue 2, pp. 143-153.

- [23] Estevez Diego Alvarez et al. (3-6 Sept. 2009) 'A continuous evaluation of the awake sleep state using fuzzy reasoning' in: Proceedings of Annual International Conference of IEEE on Engineering in Medicine and Biology Society, EMBC, pp.5539-5542.
- [24] Estevez Diego Alvarez et al. (April 2012) 'A method for the automatic analysis of the sleep macrostructure in continuum' Elsevier: Expert Systems with Applications, vol. 40, Issue 5, pp. 1796–1803.
- [25] Estevez Diego Alvarez et al.(May 2009) 'Fuzzy reasoning used to detect apneic events in the sleep apnea-hypopnea syndrome' Elsevier: Expert Systems with Applications, vol. 36, Issue 4, pp. 7778–7785.
- [26] Estevez Diego Alvarez, Bonillo V. Moret (2009) 'Model Comparison for the Detection of EEG Arousals in Sleep Apnea Patients' Springer: Bio-Inspired Systems-Computational and Ambient Intelligence, vol. 5517, pp. 997-1004.
- [27] Flores A.E. et al. (Feb 2000) 'Pattern Recognition of Sleep in Rodents Using Piezoelectric Signals Generated by Gross Body Movements' IEEE: Transactions on Biomedical Engineering, vol.54, Issue 2, pp.225-233.
- [28] Gabran S. R. I. et al. (August 20-24,2008) 'Real-time Automated Neural-Network Sleep Classifier Using Single Channel EEG Recording for Detection of Narcolepsy Episodes' in: Proceedings of 30th Annual International IEEE, EMBS Conference Vancouver, British Columbia, Canada.
- [29] Gabran S. R. I. et al. (September 2-6,2009) 'Portable Real-time Support-Vector-Machine-Based Automated Diagnosis and Detection Device of Narcolepsy Episodes' in: Proceedings of 31st Annual International Conference of the IEEE EMBS.
- [30] Golz et al. (2001) 'Application of Vector-Based Neural Networks for the Recognition of Beginning Microsleep Episodes with an Eyetracking System' in: Proceedings on the Computational Intelligence: Methods and Applications (CIMA), pp.130-134.
- [31] Gorur Dilan et al. (2002) 'Sleep Spindles Detecton using Short Time Fourier Transform and Neural Networks' in: Proceedings of IEEE 2002 International Joint Conference on Neural Networks, vol.2, pp.1631-1636.
- [32] Guimaraes G. et al. (2001) 'A method for automated temporal knowledge acquisition applied to sleeprelaed breathing disorders' artificial Intelligence in Medicine, vol. 23, pp. 211-237.
- [33] Han G. Jo et al. (July 2010) 'Genetic fuzzy classifier for sleep stage identification' Computers in Biology and Medicine, vol. 40, Issue 7, pp. 629-634.
- [34] Hassaan Amr A., Ahmed A. Morsy (August 20-24, 2008) 'Adaptive Hybrid System for Automatic Sleep Staging' in:30th Annual International IEEE EMBS Conference Vancouver, British Columbia, Canada, 2008, August 20-24.

- [35] Heiss J. E. et al. (Sept.-Oct. 2002) 'Classification of sleep stages in infants: A Neuro-Fuzzy Approach' IEEE: Engineering in Medicine and Biology Magazine, vol.21, Issue 5, pp.147-151.
- [36] Herscovici Sarah et al. (2007) 'Detecting REM sleep from the finger: an automatic REM sleep algorithm based on peripheral arterial tone (PAT) and actigraphy' Physiological Measurement., vol. 28, Issue 2.
- [37] Ho Viet Lam and Nguyen Thi My Ding 'data mining' available at http://www.ustudy.in/node/6653 (accessed on 7 March 2013).
- [38] Hsu Yeh-Liang et al. (10-12 Oct.,2005) 'Development of a portable device for home monitoring of snoring' Systems, in: Proceedings of IEEE International Conference on Man and Cybernetics, vol.3, pp.2420,2424.
- [39] Huang Liyu, Cheng Qixin Sun I Jingzhi (September 17-21, 2003)'Novel Method of Fast Automated Discrimination of Sleep Stages' in: Proceedings of the 25th Annual International Conference of the IEEE EMBS, Cancun, Mexico.
- [40] Huang Yi-Chao, Liao Min-Chuan, "The Application of a Neural Network on the Construction of the Health Management System of Hospital Staff", Journal of Quality, vol. 18, Issue. 4, pp.315-331.
- [41] Ian Watson and Farhi Marir 'case-based reasoning' available at http: // www.ai-cbr.org/classroom/cbrreview.html (accessed on 7 March 2013).
- [42] Ieong Chio-In et al. (2011) 'A Snoring Classifier based on Heart Rate Variability Analysis' Computing in Cardiology, vol. 38, pp.345–348.
- [43] Jank R. et al. (September 17-21,2003) 'Automatic Snoring Signal Analysis in Sleep Studies' in: Proceedings of the 25th Annual International Conference of the IEEE, EMBS Cancun, Mexico.
- [44] Jerome H. Friedman (1998) 'Data Mining and Statistics: What's the Connection?' Computing Science and Statistics.
- [45] Khan A. S., Hoffmann A. (2003) 'Building a casebased diet recommendation system without a knowledge engineer' Artificial Intelligence in Medicine, vol. 27, 2003, pp. 155–179.
- [46] Khandoker Ahsan H. (January 2009) 'Automated recognition of patients with obstructive sleep apnoea using wavelet-based features of electrocardiogram recordings' Journal Computers in Biology and Medicine, vol. 39, Issue 1, pp. 88-96.
- [47] Khasawneh Natheer et al. (2012) 'Combining decision trees classifiers: a case study of automatic sleep stage scoring' International Journal of Knowledge Engineering and Data Mining(IJKEDM), vol. 2, Issue 1, pp. 60-75.
- [48] Kim B.Y., Park K. S. (2000) 'Automatic sleep stage scoring system using genetic algorithms and neural network' in: Proceedings of the 22nd Annual EMBS International Conference on Engineering in Medicine and Biology Society of the IEEE, vol.2, pp.849-850.

- [49] Klink M., Quan S. F. (Apr 1987) 'Prevalence of reported sleep disturbances in a general adultpopulation and their relationship to obstructive airways diseases' Chest, vol. 91, pp. 540-546.
- [50] Koley et al. (Nov.30- Dec.1,2012 'Automated detection of apnea and hypopnea events' in: Proceedings of Third International Conference on Emerging Applications of Information Technology (EAIT), pp. 85-88.
- [51] Koton P. (1988) 'Reasoning about evidence in causal explanations' in: Proceedings of the Seventh National Conference on Artificial Intelligence, AAI Press, Menlo Park, CA, pp. 256–263.
- [52] Krajewski et al. (2007) 'Using prosodic and spectral characteristics for sleepiness detection' in: Proceedings of Interspeech, pp.1841-1844.
- [53] Kump K et al. (1994) 'Assessment of the validity and utility of a sleep-symptom questionnaire, Assessment of the validity and utility of a sleep-symptom questionnaire' Am J Respir Crit Care Med., vol. 150, Issue 3, 735-741.
- [54] Kwiatkowska M. et al. (2007) 'Knowledge-based data analysis: first step toward the creation of clinical prediction rules using a new typicality measure' IEEE- Transactions on Information Technology in Biomedicine, vol. 11, Issue 6, pp. 651–660.
- [55] Kwiatkowska M., Atkins M. S. (Aug.-Sep. 2004) 'Case representation and retrieval in the diagnosis and treatment of obstructive sleep apnea: A semio-fuzzy approach' in: Proceedings of 7th Eur. Conf. Case-Reasoning, Madrid, Spain, pp. 25–35.
- [56] Liang Sheng-Fu et al. (27-30 June 2011) 'A fuzzy inference system for sleep staging' in: Proceedings of IEEE International Conference on Fuzzy Systems (FUZZ), pp.2104-2107.
- [57] Liao Wen-Hung, Lin Yu-Kai (11-14 Oct. 2009) 'Classification of non-speech human sounds: Feature selection and snoring sound analysis Systems' in: Proceedings of IEEE International Conference on System, Man and Cybernetics (SMC), pp.2695-2700.
- [58] Lin R. et al. (2006) 'A New Approach for Identifying Sleep Apnea Syndrome Using Wavelet Transform and Neural Networks' Biomedical Engineering: Applications, Basis & Communications, vol. 18, Issue 3, pp. 138-143.
- [59] Liu D et al. (2008) 'A Neural Network Method for Detection of Obstructive Sleep Apnea and Narcolepsy Based on Pupil Size and EEG' IEEE Transactions on Neural Networks, vol. 19, Issue 2, pp. 308-318.
- [60] Liu et al. (February 2008) 'A Neural Network Method for Detection of Obstructive Sleep Apnea and Narcolepsy Based on Pupil Size and EEG' IEEE Transactions on Neural Networks, vol. 19, Issue 2, pp. 308-318.
- [61] Lopez B., Plaza E. (1997) 'Case-based learning of plans and medical diagnosis goal states', Artificial Intelligence in Medicine, vol. 9, pp. 29–60.

- [62] Maali Yashar (10-15 June 2012) 'A novel partially connected cooperative parallel PSO-SVM algorithm: Study based on sleep apnea detection' in: Proceedings of IEEE Congress on Evolutionary Computation (CEC), pp.1-8.
- [63] Maali Yashar et al. (Dec 4, 2012) 'Self-Advising SVM for Sleep Apnea Classification, in: Proceedings of the Workshop on New Trends of Computational Intelligence in Health Applications' In conjunction with the 25th Australasian Joint Conference on Artificial Intelligence, Sydney, Australia, , pp. 24-33.
- [64] Maali Yashar et al. (October 2012) 'Genetic Fuzzy Approach based Sleep Apnea/Hypopnea Detection' International Journal of Machine Learning and Computing, Vol. 2, Issue 5, pp. 685-688.
- [65] Maali Yashar, Adel Al-Jumaily (2012) 'Automated detecting and classifying of sleep apnea syndrome based on genetic-SVM' International Journal of Hybrid Intelligent Systems, vol. 9, Issue 4, pp. 203-210.
- [66] Marcos J. Victor et al. (2008) 'Radial basis function classifiers to help in the diagnosis of the obstructive sleep apnoea syndrome from nocturnal oximetry' Springer: Medical & Biological Engineering & Computing, Vol. 46, Issue 4, pp. 323-332.
- [67] Mariano Cabrero-Canosa et al. (march/april 2004) 'intelligent diagnosis of sleep apnea syndrome' IEEE: Engineering in Medicine and Biology Magazine, pp. 72-81.
- [68] Matthew T (Aug 2011) 'Machine Learning for Diabetes Decision Support' Wiley.
- [69] Mendez et al. (Dec. 2009) 'Sleep Apnea Screening by Autoregressive Models From a Single ECG Lead' IEEE: Transactions on Biomedical Engineering, vol.56, Issue 12, pp.2838-2850.
- [70] Milho Isabel et al. (2000) 'Citeseer: a user-friendly development tool for medical diagnosis based on Bayesian networks' in: proceedings of the ICEIS, pp.1-5.
- [71] Montani S. et al. (2003) 'Integrating model-based decision support in a multi-modal reasoning system for managing type 1 diabetic patients' Artificial Intelligence in Medicine, Vol. 29, pp. 131–151.
- [72] Morin CM et al. (2011) 'The insomnia severity index: psychometric indicators to detect insomnia cases and evaluate treatment response' SLEEP, vol.34, Issue 5, pp. 601-608.
- [73] Mota Cristina et al. (1999) 'Independent and Simultaneous Evolution of Fuzzy Sleep Classifiers by Genetic Algorithms' GECCO-99.
- [74] Nguyen Xuan-Lan et al. (2010) 'Insomnia symptoms and CPAP compliance in OSAS patients: A descriptive study using Data Mining methods' Elsevier: Sleep Medicine, vol.11, pp.777–784.
- [75] Norman Robert G et al. (2007) 'Detection of flow limitation in obstructive sleep apnea with an artificial neural network' Physiological Measurement, vol. 28, Issue 9.

- [76] Pandey Babita, Mishra R.B. (2009) 'Knowledge and intelligent computing system in medicine' Elsevier: Computers in Biology and Medicine, vol. 39, pp. 215 - 230.
- [77] Park Hae Jeong et al. (July 23-28,2000) 'Hybrid Neural-network and Rule-based Expert System for Automatic Sleep Stage Scoring' in: Proceedings of the 22nd Annual EMBS international Conference, Chicago IL.
- [78] Parka Hae-Jeong et al. (October 2000) 'Automated Sleep Stage Scoring Using Hybrid Rule- and Case-Based Reasoning' Computers and Biomedical Research, vol. 33, Issue 5, pp. 330–349.
- [79] Pinero Pedro et al. (June 2004) 'Sleep stage classification using fuzzy sets and machine learning techniques' Elsevier: Neurocomputing, vol. 58–60, pp. 1137-1143.
- [80] Porter B.W., Bareiss E.R.(1986) 'PROTOS: an experiment in knowledge acquisition for heuristic classification tasks' in: Proceedings of the First International Meeting on Advances in Learning (IMAL), Les Arcs, France, pp. 159–174.
- [81] Prentzas Jim, Ioannis Hatzilygeroudis et al. (May 2007) 'Categorizing approaches combining rule-based and case-based reasoning' Expert Systems, Vol. 24, No. 2, pp. 97-122.
- [82] Redline Susan et al. (2000) 'Effects of Varying Approaches for Identifying Respiratory Disturbances on Sleep Apnea Assessment, Sleep Heart Health Research Group, Am. J. Respir. Crit. Care Med., vol. 161, pp. 369-374.
- [83] Riko safaric and Andreja Rojko (2006) 'Genetic Algorithm' available at http://robin2.unimb.si/predmeti/int_reg/Predavanja/Eng/3.Genetic%20 algorithm/_18.html (accessed on 7 March 2013).
- [84] Robert N. et al. (January–February 2012) 'Is Insomnia an Independent Predictor of Obstructive Sleep Apnea' JABFM, vol. 25, Issue 1, pp. 104-110.
- [85] Roberts S., Tarassenko L.(September 1992) 'New method of automated sleep quantification' Medical and Biological Engineering and Computing, vol. 30, Issue 5, pp. 509-517.
- [86] Romero Oscar Fontenla et al. (2005) 'A new method for sleep apnea classification using wavelets and feedforward neural networks' Elsevier: Artificial Intelligence in Medicine, vol. 34, pp. 65-76.
- [87] Ronald Fisher, 'Statistical Method' available at http:// www.phil.vt.edu/dmayo/PhilStatistics/Triad/Fisher%2 01955.pdf (accessed on 7 March 2013).
- [88] Roth Thomas et al. (March 2002) 'A new questionnaire to detect sleep disorders' Elsevier: Sleep Medicine, vol. 3, Issue 2, pp. 99–108.
- [89] Saat N.Z.M. et al. (2012 'The Empirical Bayes of Occurrence of the Apnea among Sleep Apnea Patients' Journal of Applied Sciences, vol. 12, pp. 279-283.
- [90] Schaltenbrand N, Lengelle It, Macher J P. (1993) 'Neural Network Model: Application to Automatic

Analysis of Human Sleep' Computers and Biomedical Research., vol. 26, pp. 151-171.

- [91] Scott McCloskey(1999-2000) 'bayesian network' available at http://www.cim.mcgill.ca/~scott/RIT/researchPaper.ht ml (accessed on 7 March 2013).
- [92] Shmie Oren et al. (15 May 2009) 'Data mining techniques for detection of sleep arousals' Journal of Neuroscience Methods, vol.179, Issue 2, pp.331-337.
- [93] Sinha R. K. (2008) 'Artificial Neural Network and Wavelet Based Automated Detection of Sleep Spindles, REM Sleep and Wake States' Springer: Journal of medical systems, vol. 32, pp. 291-300.
- [94] Sinha R. K.(2003) 'Artificial neural network detects changes in electro-encephalogram power spectrum of different sleep-wake states in an animal model of heat stress' Springer: Medical and Biological Engineering and Computing, vol. 41, pp. 595-600.
- [95] Sloboda Jennifer, Manohar Das (20-22 July 2011) 'A Simple Sleep Stage Identification Technique for Incorporation in Inexpensive Electronic Sleep Screening Devices' in: Proceedings of IEEE National Aerospace and Electronics Conference (NAECON), pp.21-24.
- [96] Sorensena Gertrud Laura et al. (August 30 -September 3, 2011), 'Detection of arousals in Parkinson's disease patients' in: Proceedings of 33rd Annual International Conference of the IEEE, EMBS Boston, USA.
- [97] Souza José Carlos et al. (Sept. 2002 'Insomnia And Hypnotic Use In Campo Grande General Population', Arq. Neuro-Psiquiatr, Brazil, vol.60, Issue. 3-B, 702-707.
- [98] Srinivasa Gopal 'Case Based reasoning' available at http://ezinearticles.com/?Case-Based-Reasoning&id= 3405015 (accessed on 7 March 2013).
- [99] Stanley J. Swierzewski ' Sleep Disorders' available at http://www.healthcommunities.com/sleepdisorders/overview-of-sleep-disorders.shtml (accessed on 7 March 2013).
- [100]Sun Lei Ming et al. (September 2011), 'A prediction model based on an artificial intelligence system for moderate to severe obstructive sleep apnea' Sleep and Breathing, vol. 15, Issue 3, pp. 317-323.
- [101] Tagluk M. Emin, Sezgin Necmettin (2010) 'A new approach for estimation of obstructive sleep apnea syndrome' Elsevier: Expert Systems with Applications, vol. 38, pp. 5346–5351.
- [102]Tian J.Y., Liu J.Q. (2005), 'Apnea Detection Based on Time Delay Neural Network' in: Proceedings of 27th Annual International Conference of IEEE on Engineering in Medicine and Biology Society (EMBS), pp.2571-2574.
- [103]Tian J.Y., Liu J.Q. (September 1-4,2005) 'Automated Sleep Staging by a Hybrid System Comprising Neural

Network and Fuzzy Rule-based Reasoning' in: Proceedings of the 2005 IEEE Engineering in Medicine and Biology 27th Annual Conference Shanghai, China.

- [104]Ventouras Errikos M. et al. (Aug.28-Sept.1, 2012) 'Performance Evaluation of an Artificial Neural Network Automatic Spindle Detection System' Engineering in Medicine and Biology Society (EMBC), in: Proceedings of Annual International Conference of the IEEE, pp. 4328-4331.
- [105]Ventouras Errikos M. et al. (2005) 'Sleep Spindle Detection Using Artificial Neural Networks Trained with Filtered Time-Domain EEG: A Feasibility Study' Comput. Meth. Prog. Bio., vol. 78, pp. 191-207.
- [106]Victor M. Lubecke, Olga Borić-Lubecke (18-22 Jan. 2009) 'Wireless Technologies in Sleep Monitoring' IEEE:Radio and Wireless Symposium, pp. 135-138.
- [107]Vijaylaxmi et al. (2-7 Jan,2012) 'Sleep Stages Classification Using WaveletTransform & Neural Network' in: Proceedings of the IEEE-EMBS International Conference on Biomedical and Health Informatics (BHI 2012) Hong Kong and Shenzhen, China.
- [108]Vinterbo S., Ohno-Machado L. (2000) 'A genetic algorithm approach to multi-disorder diagnosis' Artificial Intelligence in Medicine, vol. 18, pp. 117– 132.
- [109]Wikipedia 'Fuzzy Logic' available at http://wiki. answers.com/Q/What_ are_ the_ advantages _and _disadvantages _of_ fuzzy_logic (accessed on 7 March 2013).
- [110]Winrow CJ et al. (2009) 'Uncovering the Genetic Landscape for Multiple Sleep-Wake Traits' PLoS ONE, vol. 4, Issue 4. pp. 1-8.
- [111]Xie Baile, Minn Hlaing (May 2012) 'Real-Time Sleep Apnea Detection by Classifier Combination' IEEE Transactions on Information Technology in Biomedicine, vol. 16, Issue 3, pp. 469-477.
- [112]Yadollahi Azadeh, Moussavi Zahra (September 2-6,2009) 'Acoustic Obstructive sleep apnea detection' in: Proceedings of 31st Annual International Conference of the IEEE EMBS, Minneapolis, Minnesota, USA.
- [113]Yldiz Abdulnasir et al. (2011) 'An expert system for automated recognition of patients with obstructive sleep apnea using electrocardiogram recordings' Elsevier: Expert Systems with Applications, vol. 38, 12880–12890.
- [114]Zoubek Lukas et al. (2007), 'Feature selection for sleep/wake stages classification using data driven methods' Elsevier:Biomedical Signal Processing and Control, vol. 2, pp. 171–179.
- [115]Zurada J.M. (1992), "Introduction to Artificial Neural Systems" West Publishing Company, USA.